\documentclass[10pt]{studiamnew}
\usepackage{graphicx}
\usepackage{amsmath}
\usepackage{amssymb}
\sloppy

\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{axiom}[theorem]{Axiom}
\newtheorem{case}[theorem]{Case}
\newtheorem{condition}[theorem]{Condition}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{criterion}[theorem]{Criterion}
\newtheorem{definition}[theorem]{Definition}

\theoremstyle{definition}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{algorithm}[theorem]{Algorithm}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{example}[theorem]{Example}
\newtheorem{exercise}[theorem]{Exercise}
\newtheorem{solution}[theorem]{Solution}
\newtheorem{notation}[theorem]{Notation}

\renewcommand{\theequation}{\thesection.\arabic{equation}}
\numberwithin{equation}{section}

\begin{document}
\begin{thebibliography}{}
\bibitem {ansaric1} Ansari, Q.H., Vector equilibrium problems and vector variational inequalities, in: Vector Variational Inequalities and Vector Equilibria. Mathematical Theories, ed. by F. Giannessi (Kluwer Academic, Dordrecht/Boston/London), 2000, 1-15.
\bibitem {ansari2} Al-Homidan, S., Ansari, Q.H., Yao, J.-C., Some generalizations of Ekeland-type variational principle with applications to equilibrium problems and fixed point theory, Nonlinear Anal. 69(1)(2008), 126-139.
\bibitem{ansari}  Ansari, Q.H.,  Vectorial  form  of  Ekeland-type variational principle with applications to vector equilibrium problems and fixed point theory, J. Math. Anal. Appl. 334(2007), 561-575.
\bibitem {ansaric2} Ansari, Q.H., Ekeland's variational principle and its extensions with applications, in: Topics in Fixed Point Theory, ed. by S. Almezel, Q.H. Ansari, M.A. Khamsi (Springer, Cham/Heidelberg/New York/Dordrecht/London), 2014, 65-100.
\bibitem{araya} Araya, Y., Ekeland's variational principle and its equivalent theorems in vector optimization, J. Math. Anal. Appl. 346(2008), 9-16.
\bibitem{araya_tanaka} Araya, Y., Kimura, K., Tanaka, T., Existence of vector equilibria via Ekeland's variational principle, Taiwanese J. Math. 12(8) (2008), 1991-2000.
\bibitem{KBP2007}  Bianchi, M., Kassay, G., Pini,  R., Ekeland's principle for vector equilibrium problems , Nonlinear Anal. 66(2007), 1454-1464.
\bibitem {blum_oettli}  Blum, E., Oettli,  W., From optimization and variational inequalities to equilibrium problems, Math. Student, 63(1-4)(1994), 123-145.
\bibitem {borwein}  Borwein, J. M., Penot, J. P., Th\'{e}ra, M., Conjugate Convex Operators, J. Math. Anal. Appl., 102(1984), 399-414.
\bibitem{castellani_giuli} Castellani,  M., Giuli,  M., Ekeland’s principle for cyclically antimonotone equilibrium problems, Nonlinear Anal.: Real World Appl. 32(2016), 213-228.
\bibitem{ekeland} Ekeland, I., Sur les probl\'{e}mes variationnels, C.R. Acad. Sci.Paris 275(1972), 1057-1059.
\bibitem {finet}  Finet, C.,  Quarta, L., Troestler, C.,  Vector-valued variational principles, Nonlinear Anal. 52(2003), 197-218.
\bibitem{Gutierrez}  Guti\'{e}rrez, C., Kassay, G., Novo, V., R\'{o}denas-Pedregosa, JL., Ekeland variational principles in vector equilibrium problems, SIAM Journal on Optimization 27(4)(2017), 2405-2425.
\bibitem{grt} G\"{o}pfert, A., Riahi, H., Tammer, Chr., Zalinescu, C., Variational Methods in Partially Ordered Spaces, Springer-Verlag, New York, 2003.
\bibitem{gtz} G\"{o}pfert, A., Tammer, Chr., Zalinescu, C., On the vectorial Ekeland’s variational principle and minimal points in product spaces, Nonlinear Anal. 39(2000), 909-922.
\bibitem{zalinescu_tammer}  Khan, A., Tammer, Chr., Z\u{a}linescu, C., Set-valued optimization, An introduction with application, Springer-Verlag, Berlin Heidelberg, 2015.
\bibitem {luc} Luc, D.T., Theory of vector optimization, Springer-Verlag, Germany, 1989.
\bibitem{oettli_thera} Oettli, W., Th\'{e}ra, M., Equivalents of Ekeland’s principle, Bull. Austral. Math. Soc. 48(1993) 385-392.
\bibitem{tammer} Tammer, Chr., A generalization of Ekeland's variational principle, Optimization 25(1992), 129-141.
\bibitem{tammer2}  Gerth (Tammer), Chr., Weidner, P., Nonconvex separation theorems and some applications in vector optimization,
J. Optim. Theory Appl. 67(1990), 297-320.
\bibitem {thera} Th\'{e}ra,  M., \'{E}tude des fonctions convexes vectorielles s\'{e}micontinues, Th\`{e}se de $3^{e}$ cycle, Universit\'{e} de Pau, 1978.
\end{thebibliography}

%
\setcounter{page}{1}
\setcounter{firstpage}{1}
\setcounter{lastpage}{12}
\renewcommand{\currentvolume}{}
\renewcommand{\currentyear}{}
\renewcommand{\currentissue}{}
%
\title{A generalized Ekeland's variational principle for vector equilibria}
\author{Mihaela Miholca}
\address{``Tehnical University of Cluj-Napoca, Romania'', \\  Department of Mathematics \\
25, G. Bari\c{t}iu Street,\\
400027 Cluj-Napoca,\\
Romania}
\email{mihaela.miholca@yahoo.com}
%
\subjclass{49J35, 49K40, 49J52}
\keywords{Ekeland's variational principle, $(k_{0},K)$-lower semicontinuity, vector triangle inequality, vector equilibria}
\begin{abstract}
In this paper, we establish an Ekeland-type variational principle for vector valued bifunctions defined on complete metric spaces with values in locally convex spaces ordered by closed convex cones. The main improvement consists in widening the class of bifunctions for which the variational principle holds. In order to prove this principle, a weak notion of continuity for vector valued functions is considered, and some of its properties are presented. We also furnish an existence result for vector equilibria in absence of
convexity assumptions, passing through the existence of approximate solutions of an optimization problem.
\end{abstract}
\maketitle

\section{Introduction}

Ekeland's variational principle (see \cite{ekeland}) has many applications in nonlinear analysis
and optimization, see \cite{ansaric1,ansari2,ansari,ansaric2,araya,araya_tanaka}, \cite{KBP2007}, \cite{gtz}, \cite{tammer}, \cite{castellani_giuli} and the reference therein. Blum, Oettli \cite{blum_oettli} and Th\'{e}ra \cite{oettli_thera} showed that their existence result for a solution of an equilibrium problem is equivalent to Ekeland-type variational principle for bifunctions.
Several authors have extended the Ekeland's variational principle to the case with a vector valued bifunction taking values in an ordered vector space, see \cite{KBP2007}, \cite{ansari}, \cite{araya_tanaka}, \cite{Gutierrez}.\\
Araya et. al. \cite{araya_tanaka} established a version of Ekeland's variational principle for vector valued bifunctions, which is expressed by the existence of a strict approximate minimizer for a weak vector equilibrium problem.\\
By a weak vector equilibrium problem we understand the problem of finding $\overline{x}\in X$ such that
\begin{equation*}
f(\overline{x}, y) \notin - intK, \text{ } \text{ for all} \text{ } y \in X,
\end{equation*}
where $f:X\times X\rightarrow Y$ is a given bifunction, $(X,d)$ is a complete metric space and $(Y,K)$ is a Hausdorff topological vector space, ordered by the closed convex cone $K$. Recall that $K\subseteq Y$ is said to be closed and convex cone if $K$ is closed, $\alpha K\subseteq K$ for all $\alpha > 0$ and $K+K \subseteq K.$\\
The approach given in Araya et. al. \cite{araya_tanaka} is based on the assumption that the equilibrium bifunction $f$ satisfies the following triangle property:
\begin{equation}\label{ti}
f(x,y)+f(y,z)\in f(x,z)+K, \text{ } \text{ for all}\text{ }  x,y,z \in X.
\end{equation}
We stress the fact that (\ref{ti}) is a rather strong condition and it is rarely verified when the equilibrium problem is a variational inequality, see \cite{castellani_giuli}.\\
Motivated and inspired by \cite{castellani_giuli}, in this paper we shall give an improvement of Theorem 2.1 in Araya et. al. \cite{araya_tanaka}. We widen the class of the vector bifunctions for which the Ekeland's variational principle is applicable. Further, some sufficient conditions for existence of equilibria which do not involve any convexity concept, neither for the domain nor for the bifunction are given, under a relaxed continuity concept for the vector functions.\\
The rest of the paper is organized as follows. In Section 2 we collect some definitions and results needed for further investigations. A weak notion of continuity for the vector valued functions is also studied and some of its properties are presented. Sections 3 and 4 are devoted to Ekeland's principles for the vector valued functions and bifunctions. Section 5 is devoted to an existence result for the weak vector equilibria where the vector bifunctions satisfy a property which generalizes the triangle inequality.
\section{Preliminaries}
Throughout this paper, unless otherwise specified, we assume that $(X,d)$ is a complete metric space, $(Y,K)$ is a locally convex Hausdorff topological vector space ordered by the nontrivial closed convex cone $K\subseteq Y$ with $int K\neq \emptyset,$ where $int K$ denotes the topological interior of $K,$ as follows:
\begin{equation*}
x \leq_{K} y \Leftrightarrow y-x \in K.
\end{equation*}
We agree that any cone contains the origin, according to the following definition.
\begin{definition}
The set $K\subseteq Y$ is called a cone iff $\lambda x \in K$ for all $x \in K$ and $\lambda \geq 0.$ The cone $K$ is pointed iff $K \cap(-K) = \{0\}$; proper iff $K\neq Y$ and $K\neq \{0\}$ .
\end{definition}
Let $k_{0}\in K\setminus (-K)$. The nonlinear scalarization function \cite{tammer2} (see also \cite{zalinescu_tammer}) $z_{K,k_{0}}: Y \rightarrow [-\infty, \infty]$ is defined as
\begin{equation*}
z_{K,k_{0}}(y)= inf\{r\in \mathbb{R} \mid y\in rk_{0}-K\}.
\end{equation*}
We present some properties of the scalarization function which will be used in the sequel.
\begin{lemma}\label{zalinescu_tammer}\cite{zalinescu_tammer}
For each $r \in \mathbb{R}$ and $y\in Y$, the following statements are true:
\begin{description}
\item [(i)] $z_{K,k_{0}}$ is proper;
\item [(ii)] $z_{K,k_{0}}$ is lower semicontinuous;
\item [(iii)] $z_{K,k_{0}}$ is sublinear;
\item [(iv)] $z_{K,k_{0}}$ is $K$ monotone;
\item [(v)] $z_{K,k_{0}}(y)\leq r \Leftrightarrow y \in rk_{0}-K$;
\item [(vi)] $z_{K,k_{0}}(y)> r \Leftrightarrow y \notin rk_{0}-K$;
\item [(vii)] $z_{K,k_{0}}(y) \geq r \Leftrightarrow y \notin rk_{0}-int K$;
\item [(viii)] $z_{K,k_{0}}(y) < r \Leftrightarrow y \in rk_{0}-int K$;
\item [(ix)] $z_{K,k_{0}}(y+\lambda k_{0})=z_{K,k_{0}}(y)+\lambda$, for every $y \in Y$ and $\lambda \in \mathbb{R}$.
\end{description}
\end{lemma}
As a corollary of the lemma above, G\"{o}pfert et al. \cite{grt} presented the following nonconvex separation theorem, see also \cite{zalinescu_tammer}.
\begin{lemma}\label{tammer_weidner1}\cite{grt}
Assume that $Y$ is a topological vector space, $K$ a closed solid convex and $A\subset Y$ a nonempty set such that $A\cap(-int K)=\emptyset$. Then $z_{K,k_{0}}$ is a finite valued continuous function such that
\begin{equation*}
z_{K,k_{0}}(-y)<0\leq z_{K,k_{0}}(x) \text{ } \text{ for all}\text{ }  x \in A \text{ } \text{ and} \text{ } y \in int K,
\end{equation*}
moreover $z_{K,k_{0}}(x)> 0$ for all $x \in intA$.
\end{lemma}
In the vector valued case there are several possible extensions of the scalar notion of \emph{lower semicontinuity}, see \cite{borwein}. We recall here the concept of $(k_{0}, K)$-lower semicontinuity introduced by Chr. Tammer \cite{tammer} which will be used in the sequel. This concept is weaker than the \emph{$K$-lower semicontinuity} which was introduced by Borwein et. al. \cite{borwein} (see also \cite{finet}, \cite{luc} and \cite{thera}.)
\begin{definition}\label{k}\cite{tammer}
A function $\varphi :X\longrightarrow Y$ is said to be:
\begin{description}
\item [(i)]\emph{ $(k_{0}, K)$-lower semicontinuous} if for all $r\in \mathbb{R}$, the set $\{x \in X: \varphi(x)\in rk_{0}-K\}$ is closed;
\item [(ii)] \emph{$(k_{0}, K)$-upper semicontinuous} if for all $r\in \mathbb{R}$, the set $\{x \in X: \varphi(x)\in rk_{0}+K\}$ is closed;
\item [(iii)] \emph{$(k_{0}, K)$-continuous} if it is both  $(k_{0}, K)$-lower semicontinuous as well as $(k_{0}, K)$-upper semicontinuous.
\end{description}
\end{definition}
The function $\varphi :X\longrightarrow Y$ is said to be \emph{$K$-bounded} below if there exists $\overline{y}\in Y$ such that $\varphi(X)\subseteq \overline{y}+K$.\\
In \cite{tammer}, the following assertion was proved.
\begin{lemma}\label{tammer}\cite{tammer}
\begin{description}
\item [(i)] If $\varphi$ is $(k_{0}, K)$-lower semicontinuous, then $z_{K,k_{0}}\circ \varphi$ is lower semicontinuous;
\item [(ii)] If $\varphi$ is $(k_{0}, K)$-upper semicontinuous, then $z_{K,k_{0}}\circ \varphi$ is upper semicontinuous.
\end{description}
\end{lemma}
\begin{remark}
It is well known that the sum of two $K$-lower semicontinuous mappings is not a $K$-lower semicontinuous mapping in general, see \cite{KBP2007}. Due to the following example, we can obtain a similar conclusion for the $(k_{0}, K)$-lower semicontinuity, i.e., if $\varphi :X\longrightarrow Y$ is $(k_{0}, K)$-lower semicontinuous, the function $\varphi(\cdot)-\varphi(x)$, where $x\in X$ is fixed, is not necessary $(k_{0}, K)$-lower semicontinuous.
\end{remark}
\begin{example}\label{e1}
Let us consider $X=\mathbb{R}^{2}$, $Y=\mathbb{R}^{2}$ and $K=\mathbb{R}^{2}_{+}$. Define $\varphi:X \rightarrow Y$ as:
\[
\varphi(x)=
\begin{cases}
(1,-2),  &  x_{1}>0, \text{  }  x_{2}\in \mathbb{R},\\
( {x_{1}},{x_{1}}), & x_{1}\leq0, \text{   } x_{2} \in \mathbb{R},
\end{cases}
\]
where $x=(x_{1}, x_{2}).$\\
This function is $(k_{0}, K)$-lower semicontinuous with $k_{0}=(1,1)$. Now take $x=(1,0)$. We will prove that the function $\varphi(\cdot)-\varphi(x)$ is not $(k_{0}, K)$-lower semicontinuous. Take also $r=1$ and consider the set
\begin{equation*}
L=\{y\in X: \varphi(y)-\varphi(x)\in (1,1)-K\}.
\end{equation*}
It is easy to observe that $y_{n}=(\frac{1}{n}, \frac{2}{n})\in L,$ $n\in \mathbb{N},$ and $y_{n}\rightarrow y_{0}$, where $y_{0}=(0,0).$ On the other hand,
\begin{equation*}
\varphi(y_{0})-\varphi(x)=(0,0)-(1,-2)=(-1, 2)\notin (1,1)-K.
\end{equation*}
Hence $y_{0}\notin L$, which shows that the set $L$ is not closed, i.e., the conclusion.
\end{example}
In what follows, we will furnish some properties for this kind of continuity for the vector functions.
\begin{proposition}
If $\varphi :X\longrightarrow Y$ is $(k_{0}, K)$-lower semicontinuous, then the function $-\varphi$ is $(k_{0}, K)$-upper semicontinuous.
\end{proposition}
\begin{theorem}\label{lower}
If $\varphi :X\longrightarrow Y$ is $(k_{0}, K)$-lower semicontinuous and $\varphi(X)\subset \bigcup_{t\in \mathbb{R}} \{tk_{0}\}$, then the function $\varphi(\cdot)-\varphi(x)$, where $x\in X$ is fixed, is $(k_{0}, K)$-lower semicontinuous.
\end{theorem}
\begin{proof}
Let us fix $x_{0} \in X$ and consider the function $\delta:X \rightarrow Y$ defined by $\delta (y)=\varphi(y)-\varphi(x_{0}), \text{ } y\in X$. Fix also $r \in \mathbb{R}$ and consider the set $S=\{y \in X :\varphi(y)-\varphi(x_{0}) \in rk_{0}-K\}$. We will prove that this set is closed.\\
Since $\varphi(X)\subset \bigcup_{t\in \mathbb{R}} \{tk_{0}\}$, it follows that, for $x_{0} \in X$, there exists $t_{0}\in \mathbb{R}$ such that $\varphi(x_{0})=t_{0}k_{0}$. We obtain
\begin{equation*}
S=\{y \in X :\varphi(y) \in (r+t_{0})k_{0}-K\}.
\end{equation*}
Since $r, t_{0} \in \mathbb{R}$ are fixed and $\varphi$ is $(k_{0}, K)$-lower semicontinuous, it follows the set $S$ is closed, i.e., the conclusion.
\end{proof}
\begin{corollary}
If $\varphi :X\longrightarrow Y$ is $(k_{0}, K)$-lower semicontinuous and $\varphi(X)\subset \bigcup_{t\in \mathbb{R}} \{tk_{0}\}$, then the function $\varphi(x)-\varphi(\cdot)$, where $x\in X$ is fixed, is $(k_{0}, K)$-upper semicontinuous.
\end{corollary}
\section{Ekeland's variational principle for the vector functions}
This section deals with an Ekeland's variational principle for the vector valued functions.
Inspired by the results obtained in Theorem 3.1 Araya \cite{araya}, we are able to present our result when the vector function is $(k_{0},K)$-lower semicontinuous.
\begin{theorem}\label{nonemptyinterior1}
If $\varphi :X \rightarrow Y$ is $(k_{0}, K)$-lower semicontinuous is such that
\begin{description}
\item[(i)] for each $x\in X$, there exists $\overline{y}\in Y$ such that $(\varphi(X)-\varphi(x))\bigcap(\overline{y}-int K)=\emptyset$;
\item[(ii)] $\varphi(X)\subset \bigcup_{t\in \mathbb{R}} \{tk_{0}\}$,
\end{description}
then, for every given $\varepsilon >0$ and for every $\widehat{x}\in X$ there exists $\overline{x}\in X$ such that:
\begin{description}
\item[(a)] $\varphi(\overline{x})-\varphi(\widehat{x})+\varepsilon d(\overline{x}, \widehat{x})k_{0}\in-K$;
\item[(b)] $\varphi(x)-\varphi(\overline{x})+\varepsilon d(\overline{x}, x)k_{0}\notin-K,$ \text{for every} $x\in X, x\neq \overline{x}$.
\end{description}
\end{theorem}
\begin{proof}
Let us consider the functional
\begin{equation*}
z_{K,k_{0}}: Y \rightarrow [-\infty, \infty],
\end{equation*}
defined by
\begin{equation*}
z_{K,k_{0}}(y)= inf\{r\in \mathbb{R} \mid y\in rk_{0}-K\}.
\end{equation*}	
For each $x\in X, \varepsilon >0$ consider the set
\begin{equation*}
S(x)=\{y\in X\mid \text{ } y=x \text{ } \text{or} \text{ } z_{K,k_{0}}(\varphi(y)-\varphi(x))+\varepsilon d(x,y)\leq 0 \}
\end{equation*}
It is obviously that $x\in S(x)$, therefore $S(x)\neq \emptyset $ for all $x\in X$. By Theorem \ref{lower}, since $\varphi$ is a $(k_{0},K)$-lower semicontinuous function, then the function $\delta(\cdot)=\varphi(\cdot)-\varphi(x)$, where $x\in X$ is fixed, is also $(k_{0},K)$-lower semicontinuous. From Lemma \ref{tammer} it follows that $z_{K,k_{0}}\circ \delta$ is lower semicontinuous and $d(x,y)$ is continuous, therefore $S(x)$ is closed for every $x\in X$.\\
Now we show that $z_{K,k_{0}}(\varphi(X)-\varphi(x)):=\cup_{y \in X}\{z_{K,k_{0}}(\varphi(y)-\varphi(x)\})$ is bounded from below for all $x \in X.$ By assumption $(i)$ and Lemma \ref{tammer_weidner1} we have that
\begin{equation*}
0\leq z_{K,k_{0}}(\varphi(y)-\varphi(x)-\overline{y}), \text{ } \text{for all} \text{ } y \in X.
\end{equation*}
Using $(iii)$ of Lemma \ref{zalinescu_tammer}, we get
\begin{equation*}
-\infty < -z_{K,k_{0}}(-\overline{y})< z_{K,k_{0}}(\varphi(y)-\varphi(x))\text{ } \text{for any} \text{ } y \in X,
\end{equation*}
which implies that $z_{K,k_{0}}(\varphi(X)-\varphi(x))$ is bounded from below.\\
Let define the real valued function
\begin{equation}\label{0}
v(x)=\inf_{y\in S(x)} z_{K,k_{0}}(\varphi (y)-\varphi (x)).
\end{equation}
and set $x=\widehat{x}\in X$. Since $z\circ \delta$ is bounded below, we have
\begin{equation*}
v(\widehat{x})=\inf_{y\in S(\widehat{x})} z_{K,k_{0}}(\varphi (y)-\varphi (\widehat{x}))> -\infty.
\end{equation*}
Starting from $\widehat{x}\in X,$ a sequence $x_{n}$ of points of $X$ can be defined such that $x_{n+1}\in S(x_{n})$ such that
\begin{equation*}
z_{K,k_{0}}(\varphi (x_{n+1})-\varphi (x_{n}))\leq v(x_{n})+\frac{1}{n+1}.
\end{equation*}
Let us take $y\in S(x_{n+1})\setminus \{x_{n+1}\}.$ It follows that
\begin{equation}\label{1}
z_{K,k_{0}}(\varphi (y)-\varphi (x_{n+1}))+\varepsilon d(x_{n+1},y)\leq 0.
\end{equation}
Since $x_{n+1}\in S(x_{n})$, we also have
\begin{equation}\label{2}
z_{K,k_{0}}(\varphi (x_{n+1})-\varphi (x_{n}))+\varepsilon d(x_{n+1},x_{n})\leq 0.
\end{equation}
Adding (\ref{1}) and (\ref{2}) we obtain
\begin{equation*}
z_{K,k_{0}}(\varphi (x_{n+1})-\varphi (x_{n}))+ z_{K,k_{0}}(\varphi (y)-\varphi (x_{n+1}))+\varepsilon d(x_{n+1},x_{n}) + \varepsilon d(x_{n+1},y) \leq 0.
\end{equation*}
Using the triangle inequality for the distance and taking into account that $z_{K,k_{0}}$ is sublinear, it follows that
\begin{equation*}
z_{K,k_{0}}(\varphi (y)-\varphi (x_{n}))+\varepsilon d(x_{n},y)\leq 0 \Longleftrightarrow y\in S(x_{n}).
\end{equation*}
Therefore, $y\in S(x_{n})$ implies that $S(x_{n+1})\subseteq S(x_{n})$.
In particular,
\begin{equation*}
v(x_{n+1})=\inf_{y\in S(x_{n+1})} z_{K,k_{0}}(\varphi (y)-\varphi (x_{n+1}))\geq \inf_{y\in S(x_{n})} z_{K,k_{0}}(\varphi (y)-\varphi (x_{n}))
\end{equation*}
\begin{equation*}
\geq (\inf_{y\in S(x_{n})} z_{K,k_{0}}(\varphi (y)-\varphi(x_{n}))-z_{K,k_{0}}(\varphi (x_{n+1})- \varphi (x_{n}))
\end{equation*}
\begin{equation}\label{00}
=v(x_{n})-z_{K,k_{0}}(\varphi (x_{n+1})- \varphi (x_{n}))\geq -\frac{1}{n+1}
\end{equation}
Thus, for $y\in S(x_{n+1})\setminus \{x_{n+1}\},$ from (\ref{0}), (\ref{1}) and (\ref{00}) we obtain
\begin{equation*}
\varepsilon d(x_{n+1},y)\leq - z_{K,k_{0}}(\varphi (y)-\varphi (x_{n+1}))\leq -v(x_n+1)\leq \frac{1}{n+1}\rightarrow 0 \text{ } \text{as} \text{ }n\rightarrow \infty,
\end{equation*}
which entails
\begin{equation*}
diam(S(x_{n}))\rightarrow 0 \text{ } \text{as} \text{ } n \rightarrow \infty.
\end{equation*}
Since the sets $S(x_{n})$ are closed and $S(x_{n+1})\subseteq S(x_{n})$ we obtain from this that the intersection of the sets $S(x_{n})$ is a singleton $\{\overline{x}\}$ and $S(\overline{x})=\{ \overline{x}\}$. This implies that $\overline{x} \in S(\widehat{x})$, or equivalently
\begin{equation*}
z_{K,k_{0}}(\varphi (\overline{x})-\varphi (\widehat{x}))\leq -\varepsilon d(\widehat{x},\overline {x}).
\end{equation*}
From Lemma \ref{zalinescu_tammer} $(v)$, it follows that
\begin{equation*}
\varphi (\overline{x})-\varphi (\widehat{x})+\varepsilon d(\widehat{x},\overline {x})k_{0}\in -K.
\end{equation*}
Therefore, $(a)$ holds.
Moreover, if $x\neq \overline{x}$, then $x \notin S(\overline{x})$, and we get
\begin{equation*}
z_{K,k_{0}}(\varphi (x)-\varphi (\overline{x}))>- \varepsilon d(x,\overline {x}).
\end{equation*}
Using again Lemma \ref{zalinescu_tammer} $(vi)$ we have
\begin{equation}\label{epb}
\varphi (x)-\varphi (\overline{x}) \notin - \varepsilon d(x,\overline {x})k_{0}-K, \text{ } \text{for all} \text{ } x\neq \overline{x},
\end{equation}
which is the conclusion $(b)$ of our theorem.
\end{proof}
\begin{remark}
In Araya \cite{araya}, an important assumption is
\begin{equation*}
(H) \text{ } \text{ } \{y \in X\mid  \varphi(y)-\varphi(x)+d(x,y)k_{0}\in -K\} \text{ } \text{ } \text{is closed for every} \text{ } \text{ } x\in X.
\end{equation*}
On the other hand, we use the $(k_{0}, K)$-lower semicontinuity for the function $\varphi$.\\
Before going further, we spend some time discussing on the comparison between the condition $(H)$ and the $(k_{0}, K)$-lower semicontinuity. Taking into account Example \ref{e1} we can observe that if the function $\varphi$ is $(k_{0}, K)$-lower semicontinuous, not necessary satisfies condition $(H)$.\\
However, if the function $\varphi$ satisfies the condition $(H)$ then is not necessary $(k_{0}, K)$-lower semicontinuous, as the following example shows.
\end{remark}
\begin{example}
Let $X=[0,1]$, $Y=l_{\infty}$ and $\varphi:X\rightarrow Y$ defined as
\[
\varphi(x)=\begin{cases}
(\frac{1}{x+1}, \frac{1}{x+2},..., \frac{1}{x+n},...), & \text{ } x\neq 0;\\
(2, \frac{1}{2},..., \frac{1}{n},...), & \text{ } x=0.
\end{cases}
\]
The ordering cone is $K_{l_{\infty}}=\{y \in l_{\infty}\mid y_{i}\geq 0$ for all $i \in \mathbb{N}\}$ and has nonempty interior.
Considering $k_{0}=(1, \frac{1}{2},..., \frac{1}{n},...)$ and $r=1$, by Definition \ref{k}, taking $x_{n}\rightarrow 0, x_{n}\in S$, it is easy to observe that the set
\begin{equation*}
S=\{x \in X: \varphi(x)\in rk_{0}-K\}
\end{equation*}
is not closed, $0 \notin S$. On the other hand, $\varphi$ satisfies the condition $(H)$. Concluding, no one implies the other.
\end{example}
\section{Ekeland's variational principle for the vector bifunctions}
Araya et al. \cite{araya_tanaka} obtained a vectorial version of Ekeland's variational principle for the bifunctions related to an equilibrium problem.
They used the triangle inequality in order to obtain the desired result.
Further, instead the triangle inequality property a suitable approximation from below of the bifunction $f$ is required.\\
Let $f:X\times X \rightarrow Y$ be a bifunction. Consider the following \emph{property}:
there exists $ \varphi : X \rightarrow Y$ such that
\begin{equation*}\label{P}
(P)\text{ } \text{ } \text{ }  f(x,y)\in \varphi(y)- \varphi(x)+ K \text{ } \text{for all} \text{ }  x, y \in X.
\end{equation*}
Property $(P)$ is more general than the triangle inequality:
\begin{equation*}\label{T}
(T) \text{ } \text{ } \text{ } f(x,z)+f(z,y)\in f(x,y)+K, \text{ } \text{for all} \text{ } x, y, z \in X.
\end{equation*}
Indeed, take in triangle inequality, for example, $\varphi_{\widehat{x}}=f(\widehat{x}, \cdot)$, where $\widehat{x}\in X$ is fixed, and property $(P)$ follows.\\
We illustrate that the property $(P)$ is more general than the triangle inequality considering the following example.
\begin{example}\label{EX}
Let $X=[0,1]$ and $Y=l_{\infty}$ and $f:X\times X \rightarrow Y$ defined as:
\[
f(x,y)=\begin{cases}
y(\frac{1}{2}, \frac{1}{4},..., \frac{1}{2^{n}},...), & \text{ } x\neq \frac{1}{2}, \text{ } y\neq \frac{1}{2};\\
(0, 0,..., 0,...), & \text{ } x= \frac{1}{2}, \text{ } y\neq \frac{1}{2};\\
(1-x)(\frac{1}{2}, \frac{1}{4},..., \frac{1}{2^{n}},...), & \text{ } x\neq \frac{1}{2}, \text{ } y= \frac{1}{2};\\
(\frac{1}{2}, \frac{1}{4},..., \frac{1}{2^{n}},...), & \text{ } x= \frac{1}{2}, \text{ } y= \frac{1}{2}.
\end{cases}
\]
The ordering cone is $K_{l_{\infty}}=\{y \in l_{\infty}\mid y_{i}\geq 0$ for all $i \in \mathbb{N}\}$.
The function $f$ does not satisfy the triangle inequality; take $x=1, y=\frac{1}{2}$ and $z=\frac{1}{4}$. We obtain
\begin{equation*}
f(1,\frac{1}{2})+f(\frac{1}{2},\frac{1}{4})\notin f(1,\frac{1}{4})+K.
\end{equation*}
On the other hand, there exists $\varphi :X\rightarrow Y$, namely
\[
\varphi(x)=\begin{cases}
(\frac{x}{2}, \frac{x}{4},..., \frac{x}{2^{n}},...), & \text{ } x\neq \frac{1}{2};\\
(\frac{1}{2}, \frac{1}{4},..., \frac{1}{2^{n}},...), & \text{ } x= \frac{1}{2},
\end{cases}
\]
such that the property $(P)$ is satisfied.
\end{example}
The following result extends Theorem 2.1 in \cite{araya_tanaka}.
\begin{theorem}\label{nt}
Let $f:X\times X\rightarrow Y$ and assume that
\begin{description}
\item[(i)] there exists $\varphi:X\rightarrow Y$ $(k_{0}, K)$-lower semicontinuous such that
\begin{equation*}
f(x,y)\in \varphi(y)-\varphi(x)+ K, \text{ } \text{for all} \text{ } x,y \in X;
\end{equation*}
\item[(ii)] for each $x\in X$, there exists $\overline{y}\in Y$ such that $(\varphi(X)-\varphi(x))\bigcap(\overline{y}-int K)=\emptyset$;
\item[(iii)] for each $x\in X$, $\{y \in X\mid (\varphi(y)-\varphi(x))+d(x,y)k_{0}\in -K\}$ is closed.
\end{description}
Then, for every $\varepsilon > 0$ and for every $\widehat{x}\in X$, there exists $\overline{x}\in X$ such that
\begin{description}
\item[(a)] $\varphi(\overline{x})-\varphi(\widehat{x})+\varepsilon d(\overline{x}, \widehat{x})k_{0}\in-K;$
\item[(b)] $f(\overline{x}, x)+ \varepsilon d(\overline{x}, y)k_{0}\notin -K$, for all $x\in X$, $x\neq \overline{x}$.
\end{description}
\end{theorem}
\begin{proof}
The function $\varphi$ satisfies all the assumptions of Theorem 3.1 in \cite{araya}. Then there exists $\overline{x}\in X$ such that item $(a)$ is verified. From the property $(P)$ we have
\begin{equation*}
f(\overline{x},x)-\varphi(x)+\varphi(\overline{x})\in K, \text{ } \text{for all} \text{ } x \in X,
\end{equation*}
and by item $(iii)$ of Theorem 3.1 we get
\begin{equation*}
\varphi(x)-\varphi(\overline{x})+\varepsilon d(\overline{x}, x)k_{0}\notin-K, \text{ } \text{for every} \text{ }  x\in X, x\neq \overline{x}.
\end{equation*}
Adding these two relations we obtain item $(b)$ of the theorem.
\end{proof}
\begin{remark}
We have to remark the fact that we do not need the assumption $f(x,x)=0$, see Theorem 2.1 in \cite{araya_tanaka}.
\end{remark}
We present now the following vectorial form of equilibrium version of Ekeland-type variational principle, result which extends similar results from the literature, see \cite{araya_tanaka}, \cite{KBP2007} and \cite{ansari}.
\begin{theorem}\label{GEP}
Let $f:X\times X\rightarrow Y$ such that
\begin{description}
\item[(i)] there exists $\varphi:X\rightarrow Y$ $(k_{0},K)$-lower semicontinuos such that
\begin{equation*}
f(x,y)\in \varphi(y)-\varphi(x)+ K, \text{ } \text{for all} \text{ } x,y \in X;
\end{equation*}
\item[(ii)] for each $x\in X$, there exists $\overline{y}\in Y$ such that $(\varphi(X)-\varphi(x))\bigcap(\overline{y}- intK)=\emptyset$;
\item[(iii)] $\varphi(X)\subset \bigcup_{t\in \mathbb{R}} \{tk_{0}\}$.
\end{description}
Then, for every $\varepsilon > 0$ and for every $\widehat{x}\in X$, there exists $\overline{x}\in X$ such that
\begin{description}
\item[(a)] $\varphi(\overline{x})-\varphi(\widehat{x})+\varepsilon d(\overline{x}, \widehat{x})k_{0}\in-K;$
\item[(b)] $f(\overline{x}, x)+ \varepsilon d(\overline{x}, x)k_{0}\notin -K$, for all $x\in X$, $x\neq \overline{x}$.
\end{description}
\end{theorem}
\begin{proof}
The idea of the proof is like in Theorem \ref{nt} and is based on Theorem \ref{nonemptyinterior1}.
\end{proof}
There are many cases where Theorem 2.1 \cite{araya_tanaka} cannot be applied but all the assumptions of Theorem \ref{GEP} are satisfied.
\begin{example}
Let $X=[0,2]$, $Y=\mathbb{R}^{2}$ and $f:X\times X \rightarrow Y$ defined as:
\[
f(x,y)=\begin{cases}
(y, 2y), & \text{ } x > 0, \text{ } y> 0;\\
(2-x, 0), & \text{ } x > 0, \text{ } y=0;\\
(y+2, y), & \text{ } x=0, \text{ } y>0;\\
(0, 0), & \text{ } x=0, \text{ } y=0.
\end{cases}
\]
The ordering cone of $Y$ is $K=\mathbb{R}^{2}_{+}$.
The function $f$ does not satisfy the triangle inequality; take $x=2, y=0$ and $z=1$. We obtain
\begin{equation*}
f(2,0)+f(0,1)\notin f(2,1)+K.
\end{equation*}
On the other hand, there exists $\varphi :X\rightarrow Y$, namely
\begin{equation*}
\varphi(x)=(x, 0),
\end{equation*}
such that $\varphi$ is $(k_{0}, K)$-lower semicontinuous with $k_{0}=(1,0)$. Moreover, $\varphi(X)\subset \bigcup_{t\in \mathbb{R}} \{tk_{0}\}$ and the property $(P)$ is satisfied.\\
We notice that $x=1$ is a solution for the weak equilibria.
\end{example}
\section{Existence solutions for the weak equilibria}
The settings for this section are the same like in the section before.\\
Using Theorem \ref{nonemptyinterior1}, we are able to show the nonemptiness of the solution set of the weak equilibria without any convexity requirements on the set $X$ and the function $f$, going through the existence of approximate solutions of an optimization problem.\\
The next statement provides the existence of solution of an optimization problem when the domain is compact.
\begin{theorem}\label{VEP1}
If $C$ is a nonempty compact subset of $X$, $\varphi :C \rightarrow Y$ is $(k_{0}, K)$-lower semicontinuous such that
\begin{description}
\item[(i)] for each $x\in C$, there exists $\overline{y}\in Y$ such that $(\varphi(C)-\varphi(x))\bigcap(\overline{y}- intK)=\emptyset$;
\item [(ii)] $\varphi(C)\subset \bigcup_{t\in \mathbb{R}} \{tk_{0}\}$;
\end{description}
then there exists $\overline{x}\in C$ such that $\varphi(y)-\varphi(\overline{x})\notin -intK,$ \text{for every} $y \in C.$
\end{theorem}
\begin{proof}
From Theorem \ref{nonemptyinterior1}, for each $n \in \mathbb{N}$, there exists $x_{n} \in C$ such that
\begin{equation*}
\varphi(y)-\varphi(x_{n})+\frac{1}{n}d(x_{n},y)k_{0}\notin -K, \text{ } \text{ for all} \text{ }  y \in C, \text{ } y\neq x_{n}.
\end{equation*}
By Lemma \ref{zalinescu_tammer} $(vi)$, we have
\begin{equation*}
z_{K, k_{0}}(\varphi(y)-\varphi(x_{n}))+\frac{1}{n}d(x_{n},y)> 0, \text{ } \text{ for all} \text{ } y \in C, \text{ } y\neq x_{n} \text{ } \text{and} \text{ }  n \in \mathbb{N}.
\end{equation*}
Since $C$ is compact, we can choose a subsequence $\{x_{n_{k}}\}$ of $x_{n}$ such that $x_{n_{k}} \rightarrow \overline{x}\in C$ as $k \rightarrow \infty$. Then, since $\varphi(y)-\varphi(\cdot)$, where $y \in C$ is fixed, is $(k_{0},K)$-upper  semicontinuous, we obtain that $z_{K, k_{0}}(\varphi(y)-\varphi(\cdot))$ is upper semicontinuous, see  Lemma \ref{tammer}.\\
Hence,
\begin{equation*}
z_{K, k_{0}}(\varphi(y)-\varphi(\overline{x}))\geq \limsup_{k \rightarrow \infty}( z_{K, k_{0}}(\varphi(y)-\varphi(x_{n_{k}}))+\frac{1}{n_{k}}d(x_{n_{k}},x))\geq 0, \text{ } \text{ for all} \text{ } y \in C.
\end{equation*}
Therefore, again by Lemma \ref{zalinescu_tammer} $(vii)$, it follows
\begin{equation*}
\varphi(y)-\varphi(\overline{x})\notin - int K, \text{ } \text{ for all} \text{ } y \in C,
\end{equation*}
and thus, $\overline{x}$ is a solution for an optimization problem.
\end{proof}
The next result gives sufficient conditions for the existence of solutions when we move to the wider class of bifunctions which satisfies the property $(P)$.
\begin{theorem}
Let $C$ be a nonempty compact subset of $X$, $f:C\times C\rightarrow Y$ a bifunction which satisfies property $(P)$ with respect to  $\varphi :C \rightarrow Y$ which is $(k_{0}, K)$-lower semicontinuous. Assume that:
\begin{description}
\item[(i)] for each $x\in C$, there exists $\overline{y}\in Y$ such that $(\varphi(C)-\varphi(x))\bigcap(\overline{y}- int K)=\emptyset$;
\item [(ii)] $\varphi(C)\subset \bigcup_{t\in \mathbb{R}} \{tk_{0}\}$,
\end{description}
Then there exists $\overline{x}\in C$ such that $f(\overline{x}, y) \notin -intK,$ \text{for every} $y \in C.$
\end{theorem}
\begin{proof}
The proof is based on Theorem \ref{VEP1} taking into account the property $(P)$.
\end{proof}
\section{Concluding remarks}
In this paper, we widen the class of vector bifunctions for which Ekeland's variational principle holds and obtain a result which improves the main result in Araya et. al \cite{araya_tanaka}. In the literature, when dealing with vector equilibrium problems and the existence of their solutions, the most used assumptions are the convexity of the domain and the generalized convexity and monotonicity, together with some weak continuity assumptions of the  vector function. In this paper, we focus on conditions that do not involve any convexity concept, neither for the domain nor for the bifunction involved. Sufficient conditions for the weak vector equilibria with bifunctions which satisfy property $(P)$, in the absence of the convexity, are given for compact domains.\\
\section{Acknowledgements}
We wish to thank an anonymous referee for useful comments and suggestions that improved the presentation of the paper.\\
The research of the author was supported by a grant of Ministry of Research and Innovation, CNCS -UEFISCDI, project number PN-III-P4-ID-PCE-2016-0190, within PNCDI III.\\


\end{document}
